Author's response to reviews

Title: Difficulties associated with out-patient management of drug abusers by general practitioners. A cross sectional survey of general practitioners with and without methadone patients in the french-speaking part of Switzerland.

Authors:

Anne AP Pelet MD (Anne.Pelet@hospvd.ch)
Jacques JB Besson Prof (Jacques.Besson@inst.hospvd.ch)
Alain AP Pecoud Prof (Alain.Pecoud@hospvd.ch)
Bernard BF Favrat Dr (Bernard.Favrat@chuv.ch)

Version: 5 Date: 21 November 2005

Author's response to reviews:

Anne Pelet
Centre Saint-Martin
Rue Saint-Martin 7
1003 Lausanne
Switzerland
Fax : 0041 21 316 16 26

Lausanne, the 20th of November 2005

Manuscript: No 5419605646972778

Dear Editor,

Here is our answer.

Answer to reviewer 1: Wendee M Wechsberg

You ask if we can challenge the national policy of inclusion of team of experts: we choose to not do it, because there is actually a trend toward a national guideline for MMT. Evaluation should certainly come afterward.

We asked BIOMEDES to correct our manuscript and to revise the English.

Answer to reviewer Nat Wright:

As Wendee M Wechsberg, you asked for revision of English. We asked BIOMEDES to correct our manuscript and to revise the English.

Anne Pelet, M.D.